Author's response to reviews

Title: Molecular Characterization and Antifungal Susceptibility Profile of Aspergillus flavus Isolates Recovered from Clinical Specimens in Kuwait

Authors:

Faten A Al-Wathiqi (fa10_1987@hotmail.com)
Suhail Ahmad (suhail_ah@hsc.edu.kw)
Ziauddin Khan (zkhan@hsc.edu.kw)

Version: 2 Date: 22 September 2012

Author's response to reviews:

Dear Editor,

The statement on the ethical approval of the study has been added (shown in red font) in the manuscript. A few minor typographical corrections have also been made.

Best regards,

Z. U. Khan